Author’s response to reviews

Title: OBJECTIVE ASSESSMENT OF PHYSICAL ACTIVITY AND SEDENTARY BEHAVIOUR IN KNEE OSTEOARTHRITIS PATIENTS - BEYOND DAILY STEPS AND TOTAL SEDENTARY TIME

Authors:

Maik Sliepen (mpl.sliepen@gmail.com;mpl.sliepen@uni-muenster.de)
Elsa Mauricio (elsa.mauricio@uni-muenster.de)
Mathijs Lipperts (matthijslipperts@gmail.com)
Bernd Grimm (bp.grimm@gmail.com)
Dieter Rosenbaum (dieter.rosenbaum@ottobock.com)

Version: 2 Date: 09 Feb 2018

Author’s response to reviews:

As requested, the following editorial revisions have been carried out:

1. The membership of Dieter Rosenbaum of the editorial board of BMC Musculoskeletal Disorders has been acknowledged in the Competing Interests section of the Declarations.
2. The cover letter and response to reviewers letter have been removed as supplementary files.
3. The manuscript has been uploaded as a single, final clean version.